Author’s response to reviews

Title: Dengue NS1 antigen as a marker of severe clinical disease

Authors:

Shiran A Paranavitane (shiran_p@live.com)
Laksiri Gomes (laksiri79@gmail.com)
Achala Kamaladasa (achalajd@gmail.com)
Thiruni N Adikari (thiruniadikari@yahoo.com)
Nilanka Munasinghe (nilanka_101@ymail.com)
S D Jayaratne (laljyaratne@gmail.com)
N L Shyamali (ajanthaliyanage@gmail.com)
Graham S Ogg (graham.ogg@ndm.ox.ac.uk)
Gathsaurie Neelika Malavige (gathsaurie.malavige@ndm.ox.ac.uk)

Version: 2
Date: 28 May 2014

Author’s response to reviews:

We have included all the authors’ email addresses, line numbers to the manuscript and contributions by all authors.